Multiple Blockholders: Good or Bad News*

Sander J.J. Konijn^{a,b}, Roman Kräussl^a, Andre Lucas^{a,b,c}

(a) Department of Finance, VU University Amsterdam
 (b) Tinbergen Institute, Amsterdam
 (c) Duisenberg school of finance

Abstract

This paper looks at the impact of dispersion in blockownership on firms' stock and debt valuation. Blockholdings by multiple blockholders is a widespread phenomenon in U.S. governance. It is not clear, however, whether dispersion in ownership among different blockholders is preferable to having a more concentrated ownership structure. We show in a theoretical model that dispersed ownership may be bad for firm value as different blockholders fail to sufficiently internalize all the negative external effects of firm value diversion. We test the implications of this empirically using a large dataset that combines blockholder information, shareholder rights information, debt ratings, and financial information of U.S. firms. We find that multiple blockholdings negatively affect Tobin's Q. The negative impact is larger for less concentrated blockownership, suggesting that a concentrated ownership structure is to be preferred on average. Results are robust to controlling for blockholder type as well as proxies for shareholder rights. Our empirical findings are also confirmed if we study the impact of ownership dispersion on firm debt ratings rather than Tobin's Q.

Key words: corporate governance, ownership structure, multiple blockholders, firm value.

JEL classification: G3, G32.

^{*}Corresponding author: Andre Lucas, FEWEB/FIN, VU University Amsterdam, De Boelelaan 1105, 1081 HV Amsterdam, The Netherlands. Phone: +31 20 598 60 39; fax: +31 20 598 6020; e-mail: alucas@feweb.vu.nl

1 Introduction

Corporate governance refers to the set of mechanisms that direct and control management activities within companies. Besides formal (e.g., contracts and legal protection) and informal (e.g., reputation) arrangements, stakeholders may also rely on a large stakeholder (internal control), or a potential large instantaneous stakeholder (external control) to restrict management's discretionary power. Kang and Shivdasani (1995), for example, show that managers of poor performing Japanese firms run a greater risk to be replaced in the presence of large shareholders. Shleifer and Vishny (1997) suggest that good governance systems seem to combine some type of large stakeholder with legal protection of their rights and those of small investors.

A critical question is whether the combination of good shareholder rights with the presence of (several) large shareholders constitutes good governance from the point of view of all stakeholders. Some studies indicate that the beneficial impact of shareholders activism may be modest (Karpoff, Malatesta and Walking (1997)). In general, large shareholders may have difficulties to cover private costs associated with the provision of what essentially is a public good (monitoring). This then leads to a suboptimal level of activism from a social point of view (Shleifer and Vishny (1986), Admati, Fleiderer and Zechner (1994)). Moreover, shifting the balance of power towards one of the stakeholders of a company could enlarge agency conflicts among different stakeholders. Large shareholders might simply try to secure private benefits by misusing their increased power for preferential self-treatment at the expense of other stakeholders (LaPorta, Lopez-de-Silanes, Shleifer and Vishny (2002)). Empirical estimates by Barclay and Holderness (1989), Nenova (2003) and Dyck and Zingales (2004) confirm the possible existence of such private benefits of control.

The main interest of our paper is on the impact of concentrated versus dispersed blockholdership on firm value and debt quality. If blockholder presence produces positive externalities such as monitoring of management, a higher concentration of blockholdings allows shareholders to internalize more of the potential benefits. This provides a larger incentive to be actively involved. It also prevents possible free-riding problems in a multiple blockholder setting (Black (1990)). Instead of governance through voice, recent studies stress the potential importance of governance through exit in mitigating agency costs. Admati and Pfleiderer (2009) and Edmans (2009) indicate that prices might better reflect fudamental value as a result of informed trading by blockholders, inducing management to undertake value-enhancing activities. Edmans and Manso (2008) show that multiple blockholders have a larger, and as a result more credible incentive to trade ex-post. Dispersed blockholding could therefore be relatively more effective in terms of governance through exit.

When blockholder presence is negatively related to firm value, Maury and Pajuste (2005) indicate that the impact of dispersed versus concentrated block ownership is subject to two countervailing forces. On the one hand, all blockholders can form a coalition, share private benefits, and in effect act like a single large blockholder. On the other hand, only a subset of blockholders might be able and willing to extract private benefits, whilst the remaining blockholders simply try to prevent this through monitoring.

We emphasize in our theoretical model a third possible outcome. In a setting with multiple blockholders, an alternative option is that different blockholders try to pursue their own objectives independently. This can be due to the fact that blockholders may be unwilling or unable in practice to form the coalitions mentioned before. The lack of coalition formation may also result from a type of prisoner's dilemma: even if blockholders are more efficient in extracting private benefits in a coordinated effort (e.g., a larger board representation), they could be better off when simply not hampering each other's attempts to extract private benefits independently. Such uncoordinated actions may be even more detrimental for company value.

Empirical evidence on the effect of multiple blockholders on firm value is limited, and virtually all non-U.S. related, see for example Faccio, Lang and Young (2001) and Volpin (2002). Differences in institutional settings and the (resulting) differences in ownership structure itself make comparisons across countries difficult, if not impossible (LaPorta, Lopez-de-Silanes and Shleifer (1999)). Though majority ownership by a single large blockholder is not common, the U.S. blockholder ownership data reveals that the presence of multiple blockholders, especially outside blockholders, is a widespread phenomenon.¹

We contribute to the empirical literature on corporate governance by investigating the relation between concentration of blockholders and firm value. Our final data set consist of approximately 3,500 firm year observations from 1996-2001. We proxy for blockholder concentration by measures related to the Herfindahl index. In addition to blockholder type, the impact

¹See also Holderness (2009).

of blockholder structure on firm value may interact with remaining shareholder rights, including those of small shareholders. To proxy for these rights, we include governance provisions in our analysis based on the constituents of the governance index by Gompers, Ishii and Metrick (2003). We use Tobin's Q as a proxy for firm value in all our regressions.

Our empirical findings reveal a negative relation between blockholding and Tobin's Q. When looking at the interaction with blockholder concentration, low concentration is more detrimental to firm value than high concentration. The results are robust to a number of variations in the design. Given the alleged importance of outside blockholders for monitoring, we also concentrate on outside blockholdings only. Again, we find a strong negative association between outside blockholding and Q. Including shareholder rights proxies in the regressions does not change the results. Compromising on shareholder rights negatively impacts Tobin's Q, but we find no significant, strong interaction effect of (outside) blockholder presence and shareholder rights.

Blockholder structure may, however, not only impact firm value, but also have a direct redistributional impact on the claim values of the firm. Thus, we investigate whether concentration of block ownership not only affects firm value, but also the value of separate debt and equity claims. We concentrate on the firms' credit ratings as a proxy of debt quality. Standard & Poor's (2005) claim that governance issues are regularly examined as part of the credit ratings process. They note that the existence of more than one owner may lead to conflicts of control. Several other studies have studied the relation between corporate governance and credit ratings, though none of them explicitly looked at blockholder concentration. Bhoraj and Sengupta (2003) and Ashbaugh-Skaife, Collins and LaFond (2006) indicate that bond ratings may be positively related to the percentage of shares held by institutional investors. While Bhoraj and Sengupta (2003) find a negative relation between the percentage of shares held by institutional blockholders and corporate bond ratings, Ashbaugh-Skaife et al. (2006) find a negative relation between the number of blockholders and ratings. Our empirical findings confirm previous findings and indicate that, whilst the presence and percentage of shares held by blockholders is important, blockholder concentration itself is relevant as well. We show that dispersed block ownership is correlated with lower rating assignments.

The remainder of this paper is organized as follows. Section 2 presents our theoretical set-up. Section 3 describes the data. Section 4 discusses our empirical results. Finally, Section 5 concludes.

2 The Model

The presence of large shareholders can have either a positive or negative effect on firm value. The *shared benefits hypothesis* suggests that a large shareholder empowered with sufficient shareholder rights is beneficial to all the company's stakeholders as he mitigates the agency problem between management and stakeholders as a group. Empirical studies on the role of shareholder activism indicate its role is modest (Karpoff, Malatesta and Walking (1996)). The *private benefits hypothesis*, by contrast, states that large shareholders can be detrimental to firm value. If blockholders pursue their own objectives, they can expropriate value from other stakeholders, such as minority shareholders, debtholders, employees, and customers. The classic agency problem between management and shareholders is replaced by an agency problem between powerful blockholders and other stakeholders of the firm (LaPorta et al. (2002)).

Empirical evidence on the possible magnitude of potential private benefits of control is sparse. Private benefit extraction is difficult to measure. Dyck and Zingales (2004) measure these benefits using an event study methodology. Their estimates range from -4 percent in Japan to +65 percent in Brazil, while their U.S. estimate equals 2.7 percent. The large variation in the estimates reveal that the value of a given block depends on other factors as well, like the company's ownership structure (e.g. number of blockholders) and the presence or absence of certain governance provisions (i.e. shareholder rights).

In this section, we use the framework of LaPorta et al. (2002) and Maury and Pajuste (2005) to study the possible effects of multiple blockholders on firm value. Let firm value be denoted by I and assume there are two blockholders. Define α_1 as the share in the residual claim held by the blockholder 1. To ensure that value diversion is inefficient, the blockholder bears a cost $c_1(s, k, \cdot)I$ when a share s is extracted (i.e. the *stealing* case). This cost explicitly depends on s and k. The variable k refers to shareholder rights, where larger values of k represent stronger shareholder rights. We assume $c_{1s} > 0$ and $c_{1ss} > 0$, such that it becomes increasingly more expensive to engage in private benefit extraction. We also assume $c_{1k} < 0$ and $c_{1ks} < 0$, implying the marginal cost of stealing decreases when shareholder rights increase. Besides blockholders interested in private benefit extraction, we allow for a blockholder not engaged in resource diversion. He holds a share α_2 in the residual claim and may want to prevent private benefit extraction by actively monitoring the actions of the first blockholder. If he does, we assume lost resources are fully recovered with probability p. We assume that monitoring can be done without any costs. Thus, the values of both blocks are given by

$$V_{1} = (1-p)(\alpha_{1}\phi_{s}I + sI - c_{1}(s,k,\cdot)I) + p(\alpha_{1}\phi_{ns}I - c_{1}(s,k,\cdot)I)$$
(1)

$$\approx (1-p)(\alpha_{1}(\phi_{ns} - \delta s)I + sI - c_{1}(s,k,\cdot)I) + p(\alpha_{1}\phi_{ns}I - c_{1}(s,k,\cdot)I)$$
(2)

$$\approx (\alpha_{1}(\phi_{ns} - q\delta s) + qs - c_{1}(s,k,\cdot))I,$$

and

$$V_2 \approx \alpha_2 (\phi_{ns} - q\delta s) I, \qquad (2)$$

where q = 1 - p, and where ϕ_s and ϕ_{ns} denote equity as a fraction of total company value in respectively the stealing and non-stealing cases. The second (first) term in the first line of (1) represents the payoff when lost resources are (not) fully recovered. The second line of (1) uses the contingent claims setting of Merton (1974), denoting the change in company value as the fraction diverted, s, times the equity delta, δ . We can interpret the term preceding Iin the last line as the *effective share* owned by the blockholder. We assume that the purpose of private benefit extraction is to maximize this effective share, possibly at the expense of other stakeholders.

Total firm value equals

$$V_{Firm} = q(1-s)I + pI.$$
(3)

A larger s thus leads to a lower firm value. Solving the first order condition (foc) of V_1 with respect to s, we obtain

$$c_{1s}(s,k,\cdot) = (1 - \delta\alpha_1)q. \tag{4}$$

Differentiating this once more with respect to α , k, and p, we get

$$\frac{ds^*}{d\alpha_1} = -\frac{(1-p)\delta}{c_{1ss}(s^*, k, \cdot)} < 0,$$
(5)

$$\frac{ds^*}{dk} = -\frac{c_{1ks}(s^*, k, \cdot)}{c_{1ss}(s^*, k, \cdot)} > 0,$$
(6)

$$\frac{ds^*}{dp} = -\frac{(1-\delta\alpha_1)}{c_{1ss}(s^*,k,\cdot)} < 0.$$
(7)

Consider initially the case of one blockholder who, by definition, is not monitored (p = 0). The first derivative reveals that a higher residual claim α_1 makes the blockholder internalize more of the value diversion effects, thus lowering his incentives to engage in value diversion. The second inequality states that an improvement in shareholder rights k, which we assume predominantly results in increased power of active shareholders, leads to better opportunities to divert resources. Finally, allowing for a second monitoring blockholder as well, a higher probability of detection lowers the optimal level of value diversion.

In a setting with two blockholders, we can distinguish three different cases: (i) blockholders collude to divert resources, maximizing their combined effective ownership stake; (ii) one blockholder diverts, while the other tries to prevent this through monitoring; (iii) both blockholders divert independently. Maury and Pajuste (2004, 2005) restrict their attention to cases (i) and (ii).

Though collusion is an option, it may be hard to achieve in practice. In addition, it is not a priori clear why it would be economically uninteresting for blockholders to divert resources independently, and to reciprocally tolerate resource diversion by the other party. We therefore also investigate alternative (iii). In case (iii), the objective function for blockholder 1 equals

$$V_1^{Ind} = (\alpha_1(\phi_{ns} - \delta(s_1 + s_2)) + s_1 - c_1(s_1, k, \cdot))I,$$
(8)

where *Ind* denotes independent diversion. Interchanging subscripts 1 and 2 we have a similar expression for blockholder 2. Both equations can be solved jointly to obtain the optimal levels of diversion.

In line with Maury and Pajuste (2004), we can derive conditions that have to be satisfied for a collusion to be sustainable. In our case, the conditions read

$$\alpha_1(\phi_{ns} - \delta s_c) + \lambda_1(s_c - c_c(s_c, k, \cdot)) \ge$$

$$\alpha_1(\phi_{ns} - q\delta s_m) + qs_m - c_1(s_m, k, \cdot)$$

$$\alpha_1(\phi_{ns} - \delta s_c) + \lambda_1(s_c - c_c(s_c, k, \cdot)) \ge$$
(10)

$$\alpha_1(\phi_{ns} - \delta(s_{1,Ind} + s_{2,Ind})) + s_{1,Ind} - c_1(s_{1,Ind}, k, \cdot)$$

$$\alpha_2(\phi_{ns} - \delta s_c) + (1 - \lambda_1)(s_c - c_c(s_c, k, \cdot)) \ge \alpha_2(\phi_{ns} - q\delta s_m)$$
(11)

$$\alpha_2(\phi_{ns} - \delta s_c) + (1 - \lambda_1)(s_c - c_c(s_c, k, \cdot)) \ge$$
(12)

$$\alpha_2(\phi_{ns} - \delta(s_{1,Ind} + s_{2,Ind})) + s_{2,Ind} - c_2(s_{2,Ind}, k, \cdot)$$

where c and m refer to collusion and monitoring case, respectively, and where λ_1 denotes the share of net private benefits accruing to blockholder 1. The conditions indicate that sustainable collusion can only arise if each blockholder's effective share is larger under collusion than in both the monitoring and the independent stealing case.

To obtain more insight in the properties of these conditions, we consider an example with quadratic cost functions $c_i(s, \cdot) = \theta_i s^2$. It is clear that the split λ_1 is a key determinant of the sustainability (and possible dominance) of a collusion outcome. If λ_1 becomes too low, blockholder 1 may be better off by stealing independently, or by taking the risk of being monitored. Vice versa, if λ_1 is too large, blockholder 2 may be better off by monitoring or by stealing independently.

For illustrative purposes we take, $\phi_{ns} = 0.25, \delta = 0.58^{-2}, \theta_2 = 23, \theta_1 = 13, \theta_c = 10, \alpha_1 = 0.2, \alpha_2 = 0.1$. In particular, we assume that larger blocks are more efficient in resource diversion.

Considering monitoring versus collusion first, we subsitute the FOCs, $s_c^* = \frac{1-\delta\alpha_c}{2\theta_c}$ and $s_m^* = \frac{(1-\delta\alpha_1)q}{2\theta_1}$ into equations (9) and (11) to get upper and lower bounds on λ_1 for each level of monitoring efficiency, p. Like Maury and Pajuste (2004), at each level p there exist subdivisions, λ_1 , such that monitoring is dominated by collusion.

The regions between the two upper (lower) curved lines of Figure I show the effective ownership shares obtained by the largest (smallest) blockholder, using these possible λ_1 at different levels of monitoring efficiency, p. Logically, less efficiency in monitoring (i.e. lower p) decreases the potential private benefits share obtained by the small, and potentially monitoring, blockholder when both blockholders decide to collude, leading to a lower effective ownership stake.

Insert Figure I

The hashed straight lines in Figure I denote the effective ownership shares obtained by substituting the optimal levels related to independent diversion, $s_{1,Ind}^*$ and $s_{2,Ind}^*$ into the right hand side of equations (10) and (12).

Now suppose, for example, that the small blockholder is reasonably good at monitoring the large blockholder. Offering a share in private benefits to the

²These levels of ϕ_{ns} and δ can, for example, be obtained using the Black and Scholes model with company value 100, zero coupon debt value 150, risk free rate 0.05, σ 0.35 and t equal to 5.

small blockholder in the A region would not be feasible, for both blockholders obtain a larger effective share when they divert resources independently. On the other hand, at B the small blockholder would be better off than independent stealing. However, the large blockholder will be worse off, so it would not be optimal to offer subdivisions in the B region. The large blockholder would rather suggest a λ_1 in region C. Unless he is powerfull enough to prevent independent diversion by the small blockholder altogether, an important assumption which would else restrict us to the monitoring versus collusion setting, the small blockholder will rather divert independently.

The point here of course being that though the summed effective shares in the collusion setting, 0.092, is larger than their summed initial ownership stakes, 0.075, their combined effective share in the independent stealing case is even larger, 0.095. As a result, one of the blockholders must be made worse off when colluding. What is worse from the point of view of other stakeholders, the optimally diverted share in the collusion setting, 0.041, is smaller than the summed diverted shares in the independent diversion case, 0.055.

In sum, dividing a specific percentage of shares amongst multiple blockholders can be more detrimental in terms of firm value for several reasons. Looking at the right hand side of (4), if one blockholder diverts resources whilst the other monitors, the negative effect of a relatively small ownership stake could dominate the positive monitoring effect. However, following Maury and Pajuste (2005), if the small blockholder can only choose between monitoring and colluding, the blockholders may well decide to collude. In that case the level of diversion is similar to the single large blockholder case.

On the other hand, coordinating stakeholders' actions by means of a coalition could in itself be quite demanding. Moreover, collusion might not be sustainable to start with. Our model indicates that it may simply be more advantageous to pursue ones own objectives independently, even when it is easier to divert resources combinedly.

In our example we made the plausible assumption that the efficiency to divert resources increases in ownership stakes. An increase in α would then result in more diversion (i.e. increasing efficiency), as well as less diversion (i.e. internalizing the externality). When efficiency increases in α , but at a decreasing rate, the locus in Figure II, depicting optimal levels of diversion, s^* , at different ownership stakes may even bend backwards. As a result, the relative distribution of ownership stakes could be quite relevant as well. For example, considering ownership stakes α_1 and α_2 , increasing α_1 by a similar amount as decreasing α_2 might have a significant effect on the total level of diverted resources when blockholders divert resources independently.

Insert Figure II

3 Data

We combine data from several sources. Standard & Poor's issuer ratings are obtained from the June 2005 Standard & Poor's CREDITPRO 7.0 database. Firm-specific data are taken from COMPUSTAT, and daily stock data are obtained from the Center for Research in Security Prices (CRSP). To obtain data on governance provisions and stock ownership, we use additional data sources described further below. In the end, matching the different data sources leaves us with between 3,315 and 3,654 firm year observations, depending on whether we use credit ratings or Tobin's Q as the dependent variable. Table I provides the variable descriptions.

Insert Table I

To proxy for shareholder rights we use the indicators on the presence or absence of individual governance provisions as constructed by Gompers et al. (2003). Their GIM index is based on publications by the Investor Responsibility Research Center (IRRC). GIM is constructed using all 22 charter provisions, bylaw provisions, and other firm-level rules, plus possible coverage under six state takeover laws, as present in the IRRC data. The data are available for July 1995, February 1998, November 1999, and January 2002. To obtain values in between reporting dates we interpolate, assuming provisions do not change until the IRRC publishes new data. To check the sensitivity of our results, we perform robustness checks with two alternative governance indices. Our first alternative index, GIB, is taken from Bebchuk, Cohen and Ferrell (2005). Based on their analysis of the impact of different governance provisions and on their discussions with lawyers in leading corporate law firms, they consider only a subset of six governance provisions. Our second alternative index is the anti-takeover index (ATI) of Cremers, Nair and Wei (2007). ATI is constructed out of four rather than 22 governance provisions and focuses on the ability of management to obstruct or delay the direct interference by shareholders.

The most widely used source for ownership data is the *Compact Disclosure* (CD) database of Standard and Poor's. Dlugosz, Fahlenbrach, Gompers and Metrick (2006) show that available data on blockholding from the CD database has many mistakes and biases such as double counting. Their cleaned database focuses on companies covered by the Investor Responsibility Research Center (IRRC) and covers the period 1996-2001.³

Blockholders are shareholders owning 5% or more of a company's stock. Such blockholders are required to file their ownership stake at the SEC. The CD database allows us to distinguish between different types of shareholders: (1) Outside blockholders; (2) Employee Stock Ownership Plans (ESOPs); (3) Officers; (4) Directors; and (5) Affiliated entities. Category 1 includes all blockholders that are not part of one of the other categories. Category 2 represents the total number of shares held by ESOPs and, by definition, does not include employee shares held through non-ESOP retirement plans. Category 3 includes officers, even when they are also directors. Category 4 refers to non-officer directors. Finally, category 5 represents any blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not completely controlled, by an officer or director of the company. Categories 2 to 5 will be referred to as inside blockholders. The CD database gives type as well as percentages of shares held by individual blockholders.

Averaging over years, more than 80 percent of the companies within our sample has at least one blockholder. Moreover, 75 percent actually has at least one outside blockholder, and this fraction gradually increases over time.⁴ If we look at the percentage of shares held by different blockholder types, we observe that affiliated shareholdings per firm can be relatively large, especially when we compare this to other inside blockholder. Most companies in our sample either have one or no inside blockholder, possibly complemented by some outside blockholding. By contrast, outside blockholders are often paired by one or more outsiders. This is reflected in the average number of outside blockholders per firm, ranging from 2.1 to 2.4.

Table II reports the distribution of blockholder types using all observations. In the first line of the table we see that for most companies outside

 $^{^{3}}$ The blockholder dataset excludes companies with multiple classes of stock. This immediately implies that a distinction between voting and cash flow rights becomes less relevant.

⁴Cremers et al. (2007) note that around 63 percent of the companies within their dataset, stretching from 1991 to 1997, have at least one *institutional* blockholder.

blockholders are the most prevalent. The lower part of the table gives an overview of successor types, conditioned on largest blockholder type. For example, the last column shows that, given that the largest blockholder is an outsider, 22.9% of the observations have no other blockholder, 9.4% have a second largest inside blockholder, and 67.7% have a second largest outside blockholder. Conditioning on the same event reveals that, when there is a third blockholder, it will most likely be an outsider as well, 42.2%.

Insert Table II

Our primary focus in this paper is on the dispersion of blockholdings. One way to measure this is to consider the number of blockholders per company. This, however, discards the effect of the shareholding distribution itself. For example, a total ownership stake of 40% held by two blockholders of equal size (20%-20%) can have very different implications than a 35%-5% distribution. The latter resembles much more a case with a single dominant blockholder than one with multiple (equally powerful) blockholders.

To also account for the block sizes held by different blockholders, we use the Herfindahl index. This index is a standard summary statistic of industry concentration in the literature on industrial organization, see for example Tirole (2003). We use a scaled version of the index defined as

$$Herfindahl = \frac{[(\%Block1) + (\%Block2) + \dots + (\%Block5)]^2}{[(\%Block1)^2 + (\%Block2)^2 + \dots + (\%Block5)^2]}.$$
 (13)

Thus, if there is just 1 blockholder, Herfindahl = 1. If there are 5 blockholders with equal shareholdings, Herfindahl = 5. If by contrast one out of 5 blockholder holds, for example, 30% of the company's shares, while the others hold 5% each, Herfindahl = 2.5, which is considerably smaller than 5.

We also experimented with alternative concentration measures, including logarithmic transformations and measures based on differences in block sizes, but our results remain qualitatively robust. We therefore do not present the full results based on these alternative measures in this paper, but only briefly comment on them when discussing the empirical results.

4 Empirical Results

4.1 Firm Value: Tobin's Q

Our sample is an unbalanced panel. To determine the effect of ownership structure and shareholder rights on Tobin's Q, we use a random effect model specification, where we include time dummies as well as industry dummies based on the 2-digit Standard Industrial Classification (SIC) codes.

Besides control variables, the first model in Table III includes both GIM as a measure of shareholder rights as well as the ownership stakes held by different blockholder types.

Insert Table III

Table III indicates that the control variables enter with their expected sign, though the coefficients on age, capital intensity and capital expenditure lack statistical significance. In particular, we would expect higher profitability, Delaware incorporation, and a higher sensitivity to general market movements, which will on average lead to higher returns, to be positively related to Tobin's Q. Firm age, size, leverage and asset tangibility tend to be negatively related to Q. Though the coefficient is negative, insignificance of firm age may be due to the fact that firm maturity is already captured by the size variable. A relatively large debt burden can be negatively (e.g. riskiness, debt overhang) or positively (e.g disciplining role) related to firm value. In accordance with the results by Demsetz and Villalonga (2001) and Maury and Pajuste (2005) we obtain a significant negative relation between Tobin's Q and leverage. Firms with high asset tangibility presumably have a lower proportion of intangible assets (goodwill, human capital), which will have a negative impact on Q. Finally, though capital expenditure may be negatively related to Q (i.e. in line with the previous argument), it could as well be an important indication of investment opportunities (Daines (2001)), leading to a positive impact on Q.

Including GIB and ATI instead of GIM we find that the coefficient on ATI is not only insignificant; at -0.03 it is quite small as well given the four point scale. All indices point in the same direction though. Compromising on shareholder rights has a negative impact on Tobin's Q, which is in line with the result of Gompers et al. (2003). Moreover, inclusion of GIB confirms the finding of Bebchuk et al. (2005). Their subindex of six governance provisions is predominantly responsible for this negative association.

Looking at blockholding, the percentage of shares held by directors seems to be the only statistically significant positive association obtained, though it is only significant at the 10 percent level. Most frequently there appears to be a negative association between blockholder ownership stakes and Tobin's Q. For example, a 10 percent increase in outside blockholding decreases Q by 0.05.

The negative association between Tobin's Q and both ESOPs and outside blockholding remains statistically significant if we include dummy variables in our equation, indicating whether a particular blockholder type is present within a specific company. The coefficients on affiliated entities and directors become insignificant.

Though not reported, including both the percentage of shares held as well as the number of blockholders by blockholder type yields no significant changes as compared with the dummy specification in case of inside blockholders. This is not surprising since inside blockholders most frequently are on their own when they are present. However, the coefficient on the total percentage of shares held by outsiders decreases to -0.20 and becomes insignificant, whilst the coefficient on the number of outside blockholders, -0.03, turns out to be significant at the 5 percent level. This already suggests that it might be highly relevant whether and to what extent a specific percentage of shares is spread out across blockholders.

In the following our main interest centers on blockholder concentration. We therefore include the suggested Herfindahl concentration measure. Confining ourselves to shareholders owning more than 5 percent of a company's shares, it is clear that there can only be one blockholder when total blockholding is less than 10 percent. In that instance blockholding stakes turn out to be less than 7.5 percent in almost two-third of the cases considered. Furthermore, it might be less reasonable to compare blockholders with combined shareholdings of 15 percent with blockholders that together hold 40 percent of a company's shares. As a result we construct three brackets: small (483 cases), medium, (1,262 cases), and large (1,374 cases), with boundaries at the 10 and 25 percentile.

Model (3) in Table III confirms the broad negative relation between percentage of shares held by blockholders and Tobin's Q. Controlling for aggregate blockholding size, the next specification, Model (4), indicates that the coefficients on the Herfindahl variables are negative and significant at the 1 percent level. The coefficients on the size variables become both less negative and less significant. This suggest that lower blockholder concentration would be relatively more detrimental in terms of firm value. Though not reported, this tendency is confirmed if we look at differences between stakes held by different blockholders.⁵ The latter measure decreases when blockholder concentration gets lower. We indeed find positive significant coefficients when we use this alternative measure.

If we look at the blockholder type coefficients, and compare these with the ones obtained in (2), we observe that the coefficients may change significantly, both in magnitude and statistical significance, predominantly with respect to outside blockholding. Given the emphasis on potential loss of value due to the presence of blockholders, the consistent strong negative impact of outsiders, outside blockholder dominance, and a potential different impact across blockholder types, it might be more appropriate to confine ourselves to outside blockholder concentration.

Our previous estimation procedure is repeated in models (5) and (6) of Table III, where we replace the total percentage of shares held by all blockholders by the total percentage of shares held by outside blockholders, whilst the concentration measure now focuses exclusively on the five largest outside blockholders. The estimation results are very much in line with the ones obtained before.

Given the negative association between Tobin's Q and blockholding, the negative coefficient on GIM seems to be inconsistent with the hypothesis that shareholder rights would predominantly benefit large, and supposedly active, shareholders. In that case, we would rather expect a positive relation between shareholder right reductions and Q. On the other hand, in line with LaPorta et al. (2002), reduced shareholder rights would also compromise on the rights of other, potentially monitoring, shareholders, making them more vulnerable to private benefit extraction.

To determine whether there are possible interaction effects we add in a next step terms that interact shareholder rights with (outside) blockholder presence. Given the ordinal scale of measurement we try to discern as good as possible between companies that, in a relative sense, do not compromise much on shareholder rights and companies with a relatively high level of governance provisions. In case of GIM we differentiate between companies with an index value equal to and below 10, the median of GIM, versus those with an index value above 10. Model (7) indicates that the coefficients are

⁵In particular, $[(\% Block1 - \% Block2)^2 + (\% Block2 - \% Block3)^2 + ... + (\% Block5 - \% Block4)]^2$ scaled by the sum of the equity stakes squared.

positive which would indeed be in line with the findings by LaPorta et al. (2002).

However, using GBI or ATI in a similar way yields no consistent results. For it should at least be difficult to observe, it could well be that private benefit extraction relies more on informal than formal arrangements, such that we will not necessarily obtain a clear association between the level of governance provisions and the presence of blockholders.

The final model specification (8) of Table III includes blockholder concentration given the size of the largest (outside) blockholder, where we again use brackets to account for size differences.⁶ The estimates on blockholder concentration gradually become smaller and less significant. Given the relatively small size of the last bracket the latter finding is not surprising. The association with Tobin's Q itself is consistent with the result obtained before. Instead of a positive relation (e.g. monitoring, contestability of power) less blockholder concentration appears to be negatively related to Q.

4.2 Debt: Corporate Credit Ratings

To determine whether and to what extent the firm's ownership structure and shareholder rights affect debt holders we look at the impact of these governance elements on corporate ratings as well.

In particular, ownership structure and shareholder rights represent 2 out of 4 dimensions of the corporarate governance score (CGS) of Standard and Poor's (2002). CGSs are developed to assess corporate governance practices and policies, and the extent to which they serve the interests of the company's financial stakeholders. Standard & Poor's (2005) states that though the CGS is geared towards the equity investor's perspective, governance issues are regularly examined as part of the credit rating methodology.

Focusing on corporate credit ratings instead of Tobin's Q we change to an ordered repons framework. We will still use a random effect specification. As a result we estimate⁷:

 $^{^{6}10\%}$ is close to the median, whilst 15% is close to the 90% percentile with respect to the size of the largest outside blockholder in both the Tobin's Q and credit rating data sets.

⁷The likelihood function can be obtained along the lines of, for example, Greene (2002), pp. 689-694, who gives a derivation in a binary respons setting.

$$\sum_{i} \ln \int_{-\infty}^{\infty} \left[\prod_{t=1}^{T_i} \left\{ F(\gamma_{y_{it}} - x_{it}^T \beta - \sigma_u u_i) - F(\gamma_{y_{it}-1} - x_{it}^T \beta - \sigma_u u_i) \right\}^{c_{i,t}} \right] f(u_i) du_i$$
(14)

by the method of maximum simulated likelihood, where $m \in \{1, 2, ..., M\}$, $\gamma_0 = -\infty$ and $\gamma_M = \infty$, and $c_{i,t}$ equals 1 if firm *i* is in the sample in period *t*. For the error terms, u_i , we use the standard normal distribution.⁸

Table IV shows that the control variables enter with their expected sign. In particular, higher (market) leverage, negative earnings, the presence of subordinated debt as well as a higher exposure to (systematic) risk, as captured by beta, decreases a company's credit rating. On the other hand, a larger size, high past profitability and a higher interest coverage ratio increases a company's debt rating. As expected, the marginal impact of a higher interest coverage ratio decreases. Whilst the coefficient on the first bracket shows the largest positive impact, the sign of the coefficient related to the last bracket actually turns out to be small and insignificant.

Insert Table IV

The coefficient on current profitability has the opposite sign of what we expected. However, including it as a stand-alone variable yields a statistically significant positive impact. The negative coefficient is probably caused by the inclusion of retained earnings, which has a strong and highly significant positive impact on ratings. D_{sub} gives an indication of the quality of a company's debt structure. For example, existing debt holders may have forced (e.g. covenants) the company to issue subordinated debt in the past to prevent possible negative wealth redistributions. Moreover, the mere existence of subordinated debt may make the issuance of claims that have a higher priority in case of default more difficult. In general, a company without subordinated debt might thus have better refinancing possibilities, which could explain the negative coefficient obtained.

Consistent with Ashbaugh-Skaife et al. (2006) the estimated coefficient on GIM reveals that a higher governance index is associated with higher ratings. In fact, though not reported, the coefficients on all governance indices enter with a positive sign. Interestingly, like ATI, the impact of GIB turns out to be insignificant now, whilst the coefficient on the index constructed out

⁸Using, for example, the logistic distribution gives almost identical results.

of governance provisions not included in GIB is still significant at the one percent level. Overall, it seems that rating agencies are foremostly concerned about shifts in balance of power towards shareholders, which might be proxied best by the total set of governance provisions. Given the positive coefficients, we may conclude that compromising on shareholder rights has a positive impact on corporate credit ratings.

Higher blockholding is without exception negatively related to debt ratings. Including ownership dummies instead of shareholdings in model (2), Table IV indicates that the negative relation between ESOPs, directors and ratings is weak and insignificant, which is in line with the low to no significance obtained in model (1). The empirical distribution of shares held by ESOPs and directors looks quite similar to that of officers. A significant difference in empirical distributions therefore does not seem to be a plausible explanation of the lack of statistical significance with respect to the total percentage of shares held in the former vis-à-vis the latter case.

Estimation error may be related to two broad factors. Given the strong negative impact on other blockholdings, the presence of inside blockholders like ESOPs and directors may, depending on circumstances, serve as a countervailing force against possible expropriation of debt value by powerful shareholders. Gordon and Pound (1993), for example, examine underlying determinants of voting outcomes on shareholder-sponsored proposals to change corporate governance structures. Whilst outside blockholders tend to support shareholder-proposed initiatives, inside blockholders on average strategically align with management. Relatedly, Shivdasani (1993) shows that shareholder blocks over which managers are likely to excercise some voting control, as well as equity ownership by CEOs themselves, turn out to have a significant negative impact on the probability of receiving a hostile takeover bid. The presence of other blockholders significantly increases this probability.

On the other hand, powerful inside blockholding may well lead to entrenchment. For instance, supermajority provisions adopted by the company, or control share acquisition laws at the state level, mandate that hostile takeover offers resisted by the firm's board must be approved by a particularly large percentage of shares, typically 66.7, 75 or 85 percent. As a result, a relatively small ownership block may effectively have veto power as far as transfers of control are concerned. Indeed, Gordon and Pound (1990) show that ESOP initiation in the presence of takeover pressure, as well as ESOPs that transferred voting control away from outside shareholders towards insiders or insider aligned groups, have a negative impact on shareholder wealth.

We document that when ESOPs, or other insider aligned blockholders, strengthen entrenchment of existing management, both shareholders and bondholders may be worse off. However, if their presence partially alleviates a potential conflict of interest between debtholders and large shareholders, their presence could be beneficial from the point of view of debtholders.⁹ Such countervailing forces may well explain the large confidence bounds obtained. Overall, though it is convenient to refer to inside and outside blockholders, the estimation results on Tobin's Q and credit ratings show that the constituent blockholder types cannot be treated as one of a kind.

Not reported, including both the number of outside blockholders as well as the percentage of shares held by them, reveals that the negative effect of outside blockholding is predominanly picked up by the former variable, with the latter becoming much smaller, -0.80, and insignificant. Moreover, whilst the average number of outside blockholders given their presence ranges from 2.1 to 2.4, the coefficient related to the number of outside blockholders itself is significant and more than 4 times smaller than the dummy specification with -0.14.

Model (3) confirms the negative relation between the percentage of shares held by blockholders and ratings. Consistent with the Tobin's Q regressions of Table III, model (4) in Table IV indeed shows that less blockholder concentration, leading to a higher value of the Herfindahl index, tends to be negatively related to credit ratings. The coefficient on the medium bracket is relatively small and lacks significance though.

Given the differences in blockholder type coefficients, in particular on outside blockholding, and the dominance of outside blockholders, the remaining models of Table IV turn to specifications restricted to outside blockholding. Model (5) confirms the relatively strong negative association between outside blockholding and ratings. The interaction terms of model (6) reveals that at least part of this negative association can be ascribed to blockholder concentration, with the magnitude of the medium bracket increasing both in magnitude and significance as compared with model (4).

Looking alternatively at differences between blockholder sizes¹⁰ compro-

⁹For example, Asquith and Wizman (1990) and Warga and Welch (1993) show that, whilst shareholders may be better off, bondholders potentially experience a significant loss in value following leveraged buyout (LBO) transactions, depending on bond type (e.g. covenants, rating, maturity).

 $^{^{10}\}mathrm{See}$ footnote 5.

mises on statistical significance (not reported). However, coefficients point consistently in the same direction. Less blockholder concentration seems to be relatively more detrimental.

Though results on blockholder concentration point in the same direction, the impact of shareholder rights turns out to be negatively related to Tobin's Q, whilst they are positively related to credit ratings. In model (7) we once more include terms that interact (outside) blockholder presence with a variable indicating whether a company, in a relative sense, does not compromise much on shareholder rights (i.e. GIM lower than or equal to 10). A negative relation between debt ratings and shareholder rights might strenghten a negative association between blockholders and ratings, or vice versa. The equation shows that the estimated coefficients indeed are consistently negative, but the effect turns out to be only (just) significant in the medium bracket region. The coefficient on GIM itself hardly changes, and remains significant.

Using GIB instead, with a cut-off point at 2, yields insignificant results. Interestingly, in case of ATI, with a cut-off point at 1, we find significant negative effects at the medium and large brackets. For ATI can be considered as a more narrow proxy of a company's takeover vulnerability, the complementarity of shareholder control and takeover vulnerability is consistent with the theoretical model of Shleifer and Vishny (1986), which shows that large shareholders more easily facilitate changes in control. From an empirical point of view this finding is somewhat in line with Cremers et al. (2007), who stress the importance of interaction between the magnitude of ATI and the presence of institutional blockholders. Using interaction with GIM, GIB or ATI as an index yields no consistent effects though.

Conditioning on blockholder concentration given the size of the largest (outside) blockholder in the last equation Model (8) confirms the negative relation between blockholder concentration and ratings.

5 Conclusion

This paper determines the effect of a firm's ownership structure, in particular the dispersion in blockholdings, on firms' stock and debt valuation. Though there are some differences between blockholder types, our dominant result is that we obtain a negative relation between blockholding and Tobin's Q. At least part of this negative association can be attributed to blockholder dispersion. The results are robust to a variety of model specifications.

The results also remain qualitatively similar if we consider the impact of blockholder concentration and shareholder rights on debt holders by using corporate credit ratings as the dependent variable. Blockholdings are negatively related to credit ratings, with a relatively larger negative impact when block dispersion is higher. In contrast to the result for Q, shareholder rights are negatively related to ratings. This suggests that a shift in balance of power towards shareholders is considered as a negative signal by credit rating agencies. Just like for Tobin's Q, we find no clear interaction effects between blockholder presence and governance provisions.

A negative impact of blockholder presence suggests there may be room for private benefits of control, possibly at the expense of other stakeholders. Though there may be competing explanations, our theoretical model and empirical results show that less blockholder concentration might aggravate this problem in two ways. First, the smaller ownership stake of blockholders in control enhances their failure to internalize negative externalities. This negative effect may be stronger than the opposite positive effect of monitoring by blockholders that are not in control. Secondly, even if blockholders are aware of their mutual incentives to divert resources, they might have no economic incentive to obstruct each others attempt to extract private benefits. This may make blockholders better off compared to the monitoring or collusion case, while the combined negative impact on firm value may be larger.

References

- Admati, Anat, Paul Fleiderer, and J. Zechner, 1994, Large Shareholder Activism, Risk Sharing, and Financial Market Equilibrium, Journal of Political Economy, Vol.102, No.6, 1097-1130.
- [2] Admati, Anat, and Paul Pfleiderer, 2009, The "Wall Street Walk" and Shareholder Activism: Exit as a Form of Voice, Review of Financial Studies, forthcoming.
- [3] Ashbaugh-Skaife, Hollis, Daniel W. Collins, and Ryan LaFond, 2006, The Effects of Corporate Governance on Firms' Credit Ratings, Journal of Accounting and Economics, Vol.42, No.1, 203–243.
- [4] Asquith, Paul, and Thierry Wizman, 1990, Event Risk, Covenants, and

Bondholder Returns in Leveraged Buyouts, Journal of Financial Economics, Vol.27, No.1, 195-213.

- [5] Barclay, Michael, and Clifford Holderness, 1989, Private Benefits from Control of Public Corporations, Journal of Financial Economics, Vol.25, No.2, 371-395.
- [6] Bebchuk, Lucian A., Alma Cohen, and Allen Ferrell, 2005, What Matters in Corporate Governance?, Unpublished Manuscript (Harvard Discussion Paper).
- [7] Bhojraj, Sanjeev, and Partha Sengupta, 2003, Effect of Corporate Governance on Bond Ratings and Yields: The Role of Institutional Investors and Outside Directors, Journal of Business, Vol.76, No.3, 455-.475.
- [8] Black, B., 1990, Shareholder Passivity Reexamined, Michigan Law Review, Vol.89, No.3, 520-608.
- [9] Cremers, K.J. Martijn, Vinay B. Nair, and Chenyang Wei, 2007, Governance Mechanism and Bond Prices, Review of Financial Studies, Vol.20, No.5, 1359-.1388.
- [10] Daines, Robert, 2001, Does Delaware Law Improve Firm Value?, Journal of Financial Economics, Vol.62, No.3, 525-558.
- [11] Demsetz, Harold, and Belén Villalonga, 2001, Ownership Structure and Corporate Performance, Journal of Corporate Finance, Vol.7, No.3, 209-233.
- [12] Dlugosz, Jennifer; Rüdiger Fahlenbrach; Paul Gompers; and Andrew Metrick, 2006, Large Blocks of Stock: Prevalence, Size, and Measurement, Journal of Corporate Finance, Vol.12, No.3, 594-618.
- [13] Dyck, Alexander, and Luis Zingales, 2004, Private Benefits of Control: An International Comparison, Journal of Finance, Vol.59, No.2, 537-600.
- [14] Edmans, Alex, 2009, Blockholder Trading, Market Efficiency, and Managerial Myopia, Journal of Finance, forthcoming.
- [15] Edmans, Alex, and Gustavo Manso, 2008, Governance Through Exit and Voice: A Theory of Multiple Blockholders, Working Paper.
- [16] Faccio, Mara, Lang, Larry H. P., and Leslie Young, 2001. Dividends and Expropriation, American Economic Review, Vol.91, No.1, 54-78.

- [17] Gompers, Paul, Joy Ishii, and Andrew Metrick, 2003, Corporate Governance and Equity Prices, Quarterly Journal of Economics, Vol.118, No.1, 107-155.
- [18] Gordon, Lilli A., and John Pound, 1990, ESOPs and Corporate Control, Journal of Financial Economics, Vol.27, No. 2, 525-555.
- [19] Gordon, Lilli A., and John Pound, 1993, Information, Ownership Structure, and Shareholder Voting: Evidence from Shareholder-Sponsored Corporate Governance Proposals, Journal of Finance, Vol.48, No.2, 697-718.
- [20] Greene, William H., 2002, Econometric Analysis, 5th print, Upper Saddle River, NJ: Prentice Hall.
- [21] Holderness, Clifford, 2009, The Myth of Diffuse Ownership in the United States, Review of Financial Studies, Vol.22, No.4, 1377-1408.
- [22] Kang, Jun-Koo, and Anil Shivdasani, 1995, Firm Performance, Corporate Governance, and Top Executive Turnover in Japan, Journal of Financial Economics, Vol.38, No.1, 29-58.
- [23] Karpoff, J., P. Malatesta, and R. Walking, 1996, Corporate Governance and Shareholder Initiatives: Empirical Evidence, Journal of Financial Economics, Vol.42, No.3, 365-395.
- [24] LaPorta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, 1999, Corporate Ownership Around the World, Journal of Finance, Vol.54, No.2, 471-517.
- [25] LaPorta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert Vishny, 2002, Investor Protection and Corporate Valuation, Journal of Finance, Vol.57, No.3, 1147-1170.
- [26] Maury, Benjamin, and Anete Pajuste, 2004, Multiple Large Shareholders and Firm Value, Unpublished Manuscript.
- [27] Maury, Benjamin, and Anete Pajuste, 2005, Multiple Large Shareholders and Firm Value, Journal of Banking and Finance, Vol.29, No.7, 1813-1834.
- [28] Merton, Robert C., 1974, On the Pricing of Corporate Debt: The Risk Structure of Interest Rates, Journal of Finance, Vol.29, No.2, 449-470.

- [29] Nenova, Tatiana, 2003, The Value of Corporate Voting Rights and Control: A Cross-Country Analysis, Journal of Financial Economics, Vol.68, No.3, 325-351.
- [30] Shivdasani, Anil, 1993, Board Composition, Ownership Structure and Hostile Takeovers, Journal of Accounting and Economics, Vol.16, No.1-3, 167-198.
- [31] Shleifer, Andrei, and Robert W. Vishny, 1986, Large Shareholders and Corporate Control, Journal of Political Economy, Vol.94, No.3, 461-488.
- [32] Shleifer, Andrei, and Robert W. Vishny, 1997, A Survey of Corporate Governance, Journal of Finance, Vol.52, No.2, 737-783.
- [33] Standard & Poor's, 2002, Standard & Poor's Corporate Governance Scores: Criteria, Methodology and Definitions, New York: McGraw-Hill Companies.
- [34] Standard and Poor's, 2005, Corporate Ratings Criteria, McGraw-Hill Companies (www.standardandpoors.com)
- [35] Tirole, Jean, 2003, The Theory of Industrial Organization. Cambridge, MA: The MIT Press.
- [36] Volpin, Paolo F., 2002, Governance with Poor Investor Protection: Evidence from Top Executive Turnover in Italy, Journal of Financial Economics, Vol.64, No.1, 61-90. erratum to: Governance with Poor Investor Protection: Evidence from Top Executive Turnover in Italy, Journal of Financial Economics, Vol.65, No.1, 159-160.
- [37] Warga, Arthur, and Ivo Welch, 1993, Bondholder Losses in Leveraged Buyouts, Review of Financial Studies, Vol.6, No.4, 959-982.

Figure I : Feasible Effective Ownership Stakes Given Monitoring Ability



The figure translates possibilities to collude, when the small blockholder can only choose between monitoring or colluding with the large blockholder, in terms of effective ownership stakes. The area between the two curved lines in the upper (lower) part of the figure denotes effective ownership stakes of the large (small) blockholder given possible subdivisions of total private benefits obtained when both blockholders collude. For comparison, the hashed lines give effective ownership stakes of the two blockholders when they decide to divert resources independently. The figure reveals that the sum of the effective ownership stakes in the latter case, 0.095, is somewhat larger than what can be obtained by a coalition, 0.092. The possibility to divert resources independently implies that collusion is unsustainable.

Figure II: Changing Ownership Concentration and Level of Diversion



The figure depicts the optimal level of diversion at different ownership stakes. When blockholders become more efficient in diverting resources as their ownership stake, a, increases we have two opposing effects. On one hand, less diversion (i.e. internalizing externalities); on the other hand, more diversion (i.e. efficiency increases). It is shown that as the efficiency to divert resources increases at a decreasing rate, a given increase in a would lead to a lower increase in diverted resources when initial blockholding is higher. Depending on the parameter setting, at a certain ownership level the efficiency increase can be dominated by the incentive to internalize the externality of resource diversion. As depicted, an increase in blockholder concentration would lead to less diversion in case 2 blockholders divert resources independently. More generally, looking from the ordinate the locus should be (strictly) concave for the latter to hold.

Total Total <th< th=""><th>No.</th><th>Variable</th><th>Description Short</th><th>Number</th><th>Description</th><th>Sign</th><th>Rating</th><th>AII</th><th>Broad</th></th<>	No.	Variable	Description Short	Number	Description	Sign	Rating	AII	Broad
Image: Section for the function of Antile Sectin Antile Section for the function of Antile	Tobin's Q	Tohin's () Market Value /Book Value	(6-60-74+ mv(CRSP))/6	(hv/assets)-hv/common stock)-deferred taxes + mv(CRSP))/hv/assets)			Count	Count
1 10, Kurshi							AAA	59	59
Current Sector Current		1 ROA	Return on Assets	18/6	(Income Before Extraordinary Items)/Total Assets	+ 7		Ċ	000
 Samo State State		2 Lev 3 CapInt	Leverage Asset Tangibility	(3+34)/b 7/6	r orar Deor / Torar Assers (Gross Property, Plant and Equipment (PPE)//Total Assets	-/+	AA+ AA	07 96	230
Control Control <t< td=""><td></td><td>4 Size</td><td>Size</td><td>In(6) Incomozotice Code</td><td>log(Total Assets)</td><td></td><td>-AA-</td><td>116</td><td></td></t<>		4 Size	Size	In(6) Incomozotice Code	log(Total Assets)		-AA-	116	
Pair And And And And And And And And And And		6 CapExp	Capital Expenditure	30/6	Capital Expenditure on Property Plant and Equipment)/Total Assets	+ '+	4+	232	989
Filty Ten and of effective field of the Matrix electron and and and and and and and and and an		7 Age	Age	CRSP	log(Months since first listing CRSP) Beta coordinicants are obtained by recreasing daily commany stock returns from the current coloridar year.	,	< <	441 316	
Mat End Concentration Concocc Concocc Concentra		م ٥	ספומ		beta democraticents are outained by regressing dany duriparty such returns notin me durient daerndar year on market returns (i.e. the Witshire5000), where we require at least 200 return on Servations. To adjust for monoclosmon to Fording Affects our on the method encounded by Dimond (1920) of define and		Ł	010	
Find Endities Endities <t< td=""><td></td><td></td><td></td><td></td><td>To inclusivity inclusions trading effects we use the interious suggested by Dimision (1979), adding one leading and one lagging value of the market trettim as explanatory variables. Though it does not affect results in fine with Blume at (1908) acrows we scale heras by their cross-sectional mean</td><td>+</td><td></td><td></td><td></td></t<>					To inclusivity inclusions trading effects we use the interious suggested by Dimision (1979), adding one leading and one lagging value of the market trettim as explanatory variables. Though it does not affect results in fine with Blume at (1908) acrows we scale heras by their cross-sectional mean	+			
Rain Set CREDITIPO 71 (June 2003) Set CREDITIPO 71 (June 2003) BEB 321 Rain Environmentation Set Code 11 the firm is a francral institution (1 dip S1C code 4) E81 221 Rain Loss Code 14 the firm is a francral institution (1 dip S1C code 4) E81 231 Rain Loss Code 14 the firm is a francral institution (1 dip S1C code 4) E81 281	Rating				וכסמומי ווו וווכ שונו בומווכ כו מי (וכסס); כמכון לכמ של ככמכ בכומה כל ווכון כוכפה סכמנטומו וווכמוי		BBB+	427	1142
1 Intuity Francis Si condition Francis Francis <td< td=""><td></td><td>Rating</td><td></td><td>S&P CREDITPRO 7.0 (Ju</td><td>ne 2005)</td><td></td><td>888- 888-</td><td>391 324</td><td></td></td<>		Rating		S&P CREDITPRO 7.0 (Ju	ne 2005)		888- 888-	391 324	
5 5 Control 5 5 Control 5 5 Control 5		1 Fin/Utility	y Fin/Utility	SIC code	1 if the firm is a financial institution (1 digit SIC code 6) or a utility (2 digit SIC code 49)	+	BB+	208	643
 Desiti and section matching in the income exert extra duration functions below of prevation matching in the income exert extra duration functions below of the inclusion of the inclusi		z Lev	Leverage (Iviarkei)	(8+34)/OHOF	10g(10ta1.De0t/TITV(URSP)) Equal to 1 # and improve before authorizational itematics 0 in the analysis true flored ware			0.47	
5 main stand 000000000000000000000000000000000000		4 Dt	Subordinated Debt	80	Equal to Thi thethiconie before extraordinary rearies < 0 hit rife previous two risear years 1 if the firm has subbridinated debt		-	DEI	
1 iei Raiand Earnings 36.6 Raiand Earnings 7.134 Asst + + Bala - + + B B - - + B - - + B B - - + B B - - + B B - - + B B - - + B B -		5 margin	Operating Margin	13/12	(Operating Income Before Depreciation) / Sales-net	+	њ В	144	226
8 Sea Month		6 ret	Retained Earnings	36/6 0000	Retained Earnings / Total Asset	+	<u>а</u>	09	
9 ov, Interest Coverage (15,178)/15 Interest Coverage (15,128)/15 Interest Coverage Interest Coverage Interest Coverage Interest Coverage Interest Coverage (15,128		/ þ 8 Size	Beta Size	LHSF In(6)	volatility que to general market movement (willsmiceoucu) log(Total Assets)	• +	'n	N	
Contraction of index and truncating can on meaningulu, we first set negative values to coverage ratios 5.10 and 5.10		9 000	Interest Coverage	(15+178)/15	In line with Blume et al. (1998), as of skewness, we adjust the way interest coverage enters the model.		0000	18	18
Governance Index GIM Gin conversion conversion conversion GIB Bebchuk et al. (2003) WFDS Uses 52 charter provisions; and other firm-level rules, plus possible coverage under 6 GIB Bebchuk et al. (2003) WFDS Uses 5 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 6 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority uses 6 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 6 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 6 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Parachules Ownership MPD Uses 4 governance provisions; classified Board, Goldon Parachules, Bylaws, Charter, Supermajority Uses 4 governance of the concent para (providen Cale) Parachules WHIP Director WHIB WHIP WHIP					For a negative value is not meaningful, we first set negative values to zero and truncate the maximum value to 100. Subsequently we use 4 brackets, with boundaries at coverage ratio's 5, 10 and 20, such that nonlineartites can be picked up by the model. For example, if the way interest coverage enters our model is denoted by Σ_{β} pcov _{fit} , an interest coverage ratio of 30 would enter as: $\cos_{fit} = 5$,	+			
Connected Inter- International (II) GIM Gompers et al. (2003) (2005) WRDs wrate (2005) Uses 22 charter provisions: bylaw provisions; and other firm-level rules, plus possible coverage under 6 state takeover laws. ATI Exemens et al. (2005) WRDs Uses 6 governance provisions: Classified Board, Golden Parachutes, Bylaws, Charter, Supermajority and Poison PII (G. Gompers et al. (2003), table 1, p.112). Ownership Digosz et al. (2006) WRDs Uses 4 governance provisions: Blank, Check, Classified Board, Special Meeting and Written Consent (G. Gompers et al. (2003), table 1, p.112). Ownership Digosz et al. (2006) Employee Share Ownership Bookholder (eg. an individual, company or trus) whose voting outcome is partially influenced, but not completely controlled, by an officier or director of the company. Minitared Minitared WRDS Bookholding by Employee Share Ownership Plans. Does not include employee shares held through non-ESOP entiment Director Nens Oricer WRDS NRDS Non-FSOP entiment anno-FSOP entiment Director Nens Outside Outside blockholders WRDS Non-FSOP entiment and enter or an eact of rectors. None Outside Outside blockholders WRDS None of the other categories. None Outsider Outside blockholders <t< td=""><td></td><td></td><td></td><td></td><td>cov_{ic}≡10 and cov_{i4}≡10.</td><td></td><td></td><td></td><td></td></t<>					cov _{ic} ≡10 and cov _{i4} ≡10.				
GIB Bebchuk et al. (2005) WFDS Distantase frakework provisions: Classified Board, Golden Parachutes, Bylaws, Charter, Supermajority ATI Cremers et al. (2007) WFDS Uses 6 governance provisions: Classified Board, Golden Parachutes, Bylaws, Charter, Supermajority ATI Cremers et al. (2007) WFDS Uses 6 governance provisions: Classified Board, Special Meeting and Written Consent Ownership Diugosz et al. (2006) Field Point Uses 4 governance provisions: Classified Board, Special Meeting and Written Consent Ownership Diugosz et al. (2006) Field Point WFDS Bockholder (e.g. an individual, company or trust) whose voling outcome is partially influenced, but not completely controlled. by an officer or director of the company. Director Piendor WFDS Bockholder (e.g. an individual, company or trust) whose voling outcome is partially influenced, but not completely controlled. by an officer or director of the company. Director Piendor WFDS Bockholder (e.g. an individual, company or trust) whose voling outcome is partially influenced, but not completely controlled. by an officer or director of the company. Director Piendor WFDS Bockholder (e.g. an individual, company or trust) whose voling outcome is partially influenced, but not completely controlled. by an officer or director of the company. Outsider Officer WFDS WFDS </td <td>GOVERNANCE II</td> <td>GIM</td> <td>Gompers et al. (2003)</td> <td>WRDS</td> <td>Uses 22 charter provisions, bylaw provisions, and other firm-level rules, plus possible coverage under 6</td> <td></td> <td></td> <td></td> <td></td>	GOVERNANCE II	GIM	Gompers et al. (2003)	WRDS	Uses 22 charter provisions, bylaw provisions, and other firm-level rules, plus possible coverage under 6				
ATI Cremers et al. (2007) WBS Tend Possime stat al. (2003), table 1, p.112). Ownership Dugosz et al. (2006) Uses 4 governance provisions: Black Check, Classified Board, Special Meeting and Written Consent (cf. Gompers et al. (2003), table 1, p.112). Ownership Dugosz et al. (2006) Insiders Uses 4 governance provisions: Black Check, Classified Board, Special Meeting and Written Consent (cf. Gompers et al. (2003), table 1, p.112). Ownership Blockholder (eg. an individual, company or trust) whose voting outcome is partially influenced, but not completely controlled, by an officier or director of the company. Minilated Afriliated Tables WRDS Director Director Director Director Outsider WRDS Outsider Usside blockholders WRDS Blockholders that are not part of one of the other categories. Outsider Usside blockholders Meridiae Afrindahl Herlindahl Index (Scaled) Herlindahl Herlindahl Index (Scaled) - Herlindahl Index (Scaled) - Herlindahl -		GIB	Bebchuk et al. (2005)	WRDS	state takeover laws. Uses 6 governance provisions: Classified Board, Golden Parachutes, Bylaws, Charter, Supermajority				
Ownership Insiders Dlugosz et al. (2006) Affilated Entities ESOP Millated Entities Employee Share Ownership Plans WRDS Blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not completely controlled, by an officer or director of the company. Affilated Entities ESOP WRDS Blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not completely controlled, by an officer of the company. Director Director WRDS Blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not completely controlled, by an officer of iter company. Director Director WRDS Blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not completely controlled, by an officer of iter company. Director Director WRDS Neno-efficer directors. Outsider Outside blockholders WRDS Neno-efficer directors. Outsider Outside blockholders WRDS Blockholders that are not part of one of the other categories. Herfindahl Index (Scaled) MRDS Blockholders t + (%Block5)] ² /((%Block1) ² + + (%Block5) ² ³ + + (%Block5) ² ³		ATI	Cremers et al. (2007)	WRDS	and Poison Fill (cf. Gompers et al. (2003), table 1, p.11/2). Uses 4 governance provisions: Blank Check, Classified Board, Special Meeting and Written Consent (cf. Gompers et al. (2003), table 1, n, 11/2).				
Insures Blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not Affiliated Affiliated Entities WRDS Blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not Affiliated Affiliated Entities WRDS Blockholder (e.g. an officer or director of the company. ESOP Employee Share Ownership WRDS Blockholder, by an officer or director of the company. Director WRDS Director WRDS Officer Officer WRDS Non-eESOP entitement plans (e.g. non-ESOP 401(k) plans). Officier Officer WRDS Non-officer directors. On-officers, even when they are are also directors. Outsider Outside Outside WRDS Difficers, even when they are are also directors. Outsider Outside Unside blockholders WRDS Blockholders that are not part of one of the other categories. Herfindahl Index (Scaled) - Herfindahl = [(%Block1) + (%Block2) + + (%Block5)] ² /([%Block1) ² + + (%Block5) ² ² + + + +	Ownership	lacidore	Dlugosz et al. (2006)		•				
Affiliated Affiliated Entitiated Affiliated Entitiated Entitiated Entitiated Filter WTBS completely controlled, by an officer or director of the company. ESOP Employee Employee Share WRDS Blockholding by Employee Share Downership WRDS Director Director WRDS Non-officer Officiers, even when they are are also directors. Outsider Otiside Outside WRDS Blockholders that are not part of one of the other categories. Outsider Outside WRDS Blockholders that are not part of one of the other categories. Herfindahl Index (Scaled) Herfindahl = [(%Block1) + (%Block2) + + (%Block5)] ² ([(%Block1) ² + + (%Block5) ² ²) ¹ + + (%Block5) ² ²) ² + + (%Block5) ² ²		linsiders			Blockholder (e.g. an individual, company or trust) whose voting outcome is partially influenced, but not				
Director WRDs Non-officient directors. Officient Officient WRDs Officient Officient WRDs Outsider Officient WRDs Outsider Outside blockholders WRDs Blockholders WRDs Blockholders that are not part of one of the other categories. Herfindahl Index (Scaled) - Herfindahl = [(%Block1) + (%Block2) + + (%Block5)] ² ([(%Block1) ² + + (%Block5) ²) ²]		Affiliatec ESOP	Affiliated Entities Employee Share Ownership Plans	WRDS WRDS	completely controlled, by an officer or director of the company. Blockholding by Employee Share Ownership Plans. Does not include employee shares held through				
Outsider Outside blockholders WRDS Blockholders that are not part of one of the other categories. Outsider Outside blockholders WRDS Blockholders that are not part of one of the other categories. Herfindahl Index (Scaled) Herfindahl = [(%Block1) + (%Block2) + + (%Block5)] ² ([(%Block1) ² + + (%Block5) ²) ²)		Director	· Director Officer	WRDS WRDS	Non-officer directions, (view rest of the second				
Outsider Outside blockholders WRDS Blockholders that are not part of one of the other categories. Herfindahi Herfindahi Index (Scaled) - Herfindahi = [(%Block1) + (%Block2) + + (%Block2) + + (%Block5)] ² [(%Block1) ² + (%Block5) ² + + (%Block5) ²]		Outsideı							
Herfindahi Index (Scaled) - Herfindahi = [(%Block1) + (%Block2) + + (%Block5)] ² [(%Block1) ² + (%Block2) ² + + (%Block5) ²]		Outsider	r Outside blockholders	WRDS	Blockholders that are not part of one of the other categories.				
		Herfindal	hl Herfindahl Index (Scaled)		Herfindahl = [(%Block1) +(%Block2)+ + (%Block5)] ² /[(%Block1) ² +(%Block2) ² + + (%Block5) ²]				

Table I : Variable description

To merge the different data sets we use the following time conventions. The time period considered are the calendar years stretching from 1996 to 2001. In each specific year we take the end of the year as our point of reference. Firm specific variables constructed using COMPUSTAT data are always related to the last closed fiscal year. Variables constructed out of CRSP data are always up to date. That is, market values of equity are calculated as the number of shares outstanding times the stock price at the end of the year. Betas are estimated using daily data from the current calendar year, where we require at least 200 return observations. In line with Gompers et al. (2003), we assume governance provisions do not change until the IRRC publishes new data. We use the blockholding data reported in the specific calendar vear considered.

Table II : Blockholder Size

Notes: All statistics relate to the Tobin's Q sample of 3654 firm-year observations. The rating sample yields qualitatively similar results. The upper part of the table shows the largest blockholder type distribution, scaled by the total number of observations. The lower part of the table gives an overview of successors, conditioned on largest blockholder type. Affiliated entities, ESOPs, Directors and Officers are merged into one insider category.

			Larges	t Blockholde	r	
		Affiliated	ESOP	Director	Officer	Outsider
		%	%	%	%	%
Coverage		7.0	6.4	3.6	5.8	62.5
			Su	ccessors		
Second	None	22.2	39.7	34.1	25.5	22.9
	Insider	20.6	8.5	22.0	20.3	9.4
	Outsider	57.2	51.7	43.9	54.2	67.7
Third	None	52.5	71.4	56.1	48.6	52.2
	Insider	10.9	4.3	6.8	9.0	5.5
	Outsider	36.6	24.4	37.1	42.5	42.3
Fourth	None	77.4	87.2	73.5	69.8	75.2
	Insider	7.4	1.3	10.6	4.2	2.8
	Outsider	15.2	11.5	15.9	25.9	22.0
Fifth	None	92.6	94.0	87.1	89.2	88.9
	Insider	1.6	0.9	4.5	1.4	1.1
	Outsider	5.8	5.1	8.3	9.4	10.1

~
20
j
õ
Ŧ
g
Ř
7
2
ø
S
.9
j <u>s</u>
ō
à
D)
ğ
a
Ξ
é
5
G
<u>.</u>
Q
j,
÷
٦,
<u> </u>
1
Φ
ble
Table

		(0.22) a	(0.10) a	(0.07)	(0.02) a	(0.32) a	(0.03) (0.02) a	(0.01) a		(0.05) (0.05) b (0.06) (0.05)	(0.49) b (0.26) (0.21) c			(0.02) b (0.02) c (0.04)					
	(8)	2.18	-0.68	-0.07	-0.Ub	0.33	-0.02 0.14	-0.02		-0.05 -0.10 0.01 -0.06	-0.98 -0.30 -0.34			-0.04 -0.03 -0.02	0.20	828	0.31	0.40	
		22) a 16) a	10) a 10) a	07)	05) a	32) a	03) 02) a	01) c		05) 05) b 05) 05)	63) a 46) 19) c	03) b 02) b	06) c 05) b 06)						
	(2)	2.07 (0.	-0.68 (0.	-0.07 (0.	-0.06 (U.	0.33 (0.	-0.02 0.14 (0.	-0.01 (0.		-0.05 -0.10 0.01 -0.06 (0.	-2.07 (0. -0.11 (0.	-0.07 (0. -0.04 (0.	0.08 (0. 0.10 (0. 0.03 (0.		0.20	828	0.31	0.40	
		а (ש מ (а С	م	a)) a		٩	c a	م م (((
		8 (0.22 0.16	8 (0.10	7 (0.07	0.UZ	3 (0.32	2 (0.03 (0.02	2 (0.01		5 (0.05 0 (0.05 6 (0.05 6 (0.05	0 (0.46 4 (0.44 7 (0.18	7 (0.03 4 (0.02			0 5	+ 00	-	0	
	(6	2 1 1 2 0 0	9.0 9.0	0.0		- 0.0	0.0	-0.0		0.0 1.0 0.0 0.0	-1.5 0.2	0.0 0.0			0.2	82	0.3	0.4	
		а () ()	а а с	5	ດ ດ ໂມເມີ	ช ก (ก	ଅ ଜ ରି	1) a		م م (م) (م) (م)	0) a a 0) a a								
s		(0.22	0.10	0.0		0.3 0.3 0.3	0.0	-0.0		00000	(0.18 (0.18 (0.11				_				
Outsider:	(2)	2.18	-0.68	-0.07	90.0- 0-	0.31	-0.02 0.14	-0.02		-0.05 -0.10 -0.02 -0.06	-1.29 -0.61 -0.54				0.20	828	0.30	0.41	
-		5	វល		L DI	2	ъ	в		U	ດ ບ	0 0							
		(0.22)	(0.10)	(0.07)	(0.UZ)	(0.32)	(0.03) (0.02)	(0.01)		(0.05) (0.05) (0.06) (0.06) (0.04)	(0.64) (0.42) (0.16)	(0.03) (0.02)							
	(4)	2.18	-0.67	-0.07	-00	0.33	-0.02 0.15	-0.02		0.01 -0.04 0.08 0.00 0.00	-2.19 0.19 -0.23	-0.09			0.20	828	0.31	0.40	
		5 5 5	л а О) а	6	a a	۳ آن (آ	ы С) ЭЭ	1) a		(c) (c) (c) (c) (f	2 () () () () () () () () () () () () () (
s	_	(0.2	0.1	0.0	0.0	0.0)	0.0	(0.0			0.0							_	
All Block	(3)	2.20	-0.68	-0.07	90.0-	0.31	-0.02 0.14	-0.02		0.05 0.07 0.01 0.03	-1.39 -0.42 -0.46				0.20	828	0.31	0.40	
		.22) a 16) a	.10) a .10) a	.07)	.02) a	.32) u	.03) .02) a	.01) a		.05) b .06) b .05) a									
)	(2)	0) 06 0)	20 20 20	07 (0	2) (CO. FF	ر 38 (0 38:	.01 (0 .15 (0	.02 (0		.03 (0 .03 (0 .05 (0 .05) (0 .05 (0 .05) (0)) (0 .05) (0)) (0)) (0)) (0)) (0)) (0)) (0)) (.19	328	31	41	
		CÌ C	ιĢ	φ (ρ c	Ō	φö	Ģ		φ φ ο φ φ					0.96	5 8	C	Ö	
		(0.22) a	0.10) a	(0.07)	(0.02) a	0.32) a	(0.03) (0.02) a	(0.01) a	(0.23) b (0.42) b (0.34) c (0.37) a										
	(1)	2.12 (-0.68	-0.06	00 00	0.26	-0.02 0.14 (-0.02 (-0.38 -0.90 0.50 -0.51 -0.51						0.19	828	0.30	0.42	
											gregated	, m e	<i>er rights</i> II ium e	 est_{Small} est_{Medium} st_{Large} 					
	ble			٦t	0.01	.xp			liated OP ∋ctor icer sider	tted د tor ئر	<i>cholding ag</i> all dium ge	<i>rdahl index</i> ndahl*Medi ndahl*Largu	<i>sharehold</i> € ≤ 10)*Sma ≤ 10)*Med ≤ 10)*Larg	<i>ndahl inde.</i> ndahl*Larg ndahl*Larg ndahl*Larg					
	Varia	0 0 0	Lev	Caplr 0:	Dolou	CapE	Age B	GIM	%Afffi %ES(%Dirt %Offi	Affilia ESOF Direc Office Outsi	<i>Block</i> %Sm %Mec	<i>Herfii</i> Herfir Herfir	High (GIM (GIM (GIM	<i>Herfi</i> Herfir Herfir Herfir	2 H2	firms	, D	b	,

reported), and im dummers (not reported). GIM denotes the governance index proposed by Gompers et al. (2003), consisting of 24 provisions, which adds 1 point if a provisions compromises on shareholder rights (GIM). The remaining variables refer to a firm's ownership structure, in terms of blockholders are shareholders owning 5 percent or more of a firm's stock. Ownership variables enter either as percentiles, always preceded by %, or as 0-1 indicators. The blockholder types considered are: Affiliated entities (Affiliated), Employee Stock Ownership Plans (ESOP), non-officer directors (Diffeer) officers, even when they are also directors (Officer) and outside blockholders (Outsider). %Small, %Medium and %Large denote the total percentage of shares held by blockholders subdivided within three brackets, with boundaries at the 5, 10 and 25 percent level. The Herfindahl variable measures ownership concentration, considering the (potentially) 5 largest blockholders. Herfindahl variable This table presents random effect unbalanced panel regressions of Tobin's Q on governance and control variables for 828 U.S. companies over the period 1996-2001. To reduce the weight of outliers Tobin's Q is capped at the 5 and 95 percentiles. The control variables are capped at the 5 and 95 percentiles. The control variables are capped at the 5 and 99.5 percentiles. The control variables used are: Return on Assets (ROA), Leverage (Lev), Capital Intensity (CapInt), Firm Size (Size), a dummy equal to one if the company is incorporated in Delaware (Delaware), Capital Expenditure (CapExp), Firm Age (Age) and Market Beta (B), industry dummies based on 2 digit SIC codes (not (% Block5)² (% Block1)² + (% Block2)² +... + (% Block5)²). Largest_{smal}. Largest_{smal} and Largest_{mage} indicate whether the largest blockholder holds a block position in the intervals [5,10), [10,15), or 15 and beyond. Except for the type dummies, ownership variables at the right hand side of the table solved blockholders only. σ_v and σ_u denote respectively the estimated between-groups and within-groups variance. Standard errors are in parentheses. a,b,c denote statistical significance at respectively the 1, 5 and 10 percent level.

					All Bloc	ks				0	utsiders										
	(1)		(2	(;	;)	3)		(4)			(2)			(9)			(2)		8)	(
Fin/Utility Lev	4.36	(0.61) a (0.08) a	4 0 4 0	5 (0.44) a (0.08) a	4 C	(0.42 0.05	8 8 0 0	4.38 -0.99	(0.63) (0.08)		4.29 -0.98	(0.42) (0.08)	ით თ	5.24	(0.61) (0.08)	, , , , ,	4.55 -0.99	(0.47) a (0.08) a	4.2	8 (0.42) 8 (0.08)	m ~
loss _{(t+1,1}	-1.03	(0.28) a	-1.0	4 (0.28) a	- 1.0	0.28	3) a	-1.09	(0.28)	5 10	-1.03	(0.28)	ស	-1.02	(0.29)	ច	-0.99	(0.28) a	-1.0	3 (0.28)	
D _{sub}	-0.55	(0.22) a	-0.5	4 (0.22) a	9.Q	61 (0.22	2) b	-0.62	(0.22)	g	-0.50	(0.22)	q	-0.69	(0.24)	а	-0.66	(0.22) a	-0.5	3 (0.22)	~
coV ₁	0.30	(0.06) a	0.3	1 (0.06) a	0.3	1 (0.06	3) a	0.31	(0.06)	g	0.30	(0.06)	a	0.29	(0.06)	a	0.32	(0.06) a	0.3	1 (0.06)	m,
cov ₂	0.13	(0.04) a	0.1	4 (0.04) a	0.1	4 (0.02	4) a	0.14	(0.04)	g	0.14	(0.04)	g	0.15	(0.04)	g	0.12	(0.05) a	0.1	4 (0.04)	m,
cov ₃	0.07	(0.03) b	0.0	7 (0.03) а	0.0	0.03	3) a	0.07	(0.03)	g	0.07	(0.03)	a	0.07	(0.03)	q	0.07	(0.03) a	0.0	7 (0.03)	m
cov ₄	0.00	(0.01)	0.0	0 (0.01)	0.0	0.0	= 7	0.00	(0.01)		0.00	(0.01)	_	0.00	(0.01)	_	0.00	(0.01)	0.0	0.01)	
margin rat	-1.38	(0.7.7) b	י מימ	9 (0./4) b 7 (0.49) a		2.0. 	0 0 () (-1.49 5.76	(0.76)	Ω (-1.43 5.78	(0.70) (0.51)	م م	-1.45 7 80	(0./3)	۵ n	-1.34 5 70	(0.69) b	 	9 (0.69) 0 (0.51)	<u> </u>
B C	-0.46	(0.09) a	0 9 4	3 (0.09) a	4. Q	13 (0.0) 13 (0.0)	эе (-	-0.45	(0.09)	ຽດເ	-0.43	(0.09)	5 10	-0.41 -0.41	(0.09)	ຽດເ	-0.43	(0.09) a	4.0	(0.09)	
Size	2.54	(0.13) a	2.5	8 (0.13) a	2.6	2 (0.13	3) a	2.54	(0.13)	g	2.64	(0.13)	B	2.59	(0.14)	a	2.57	(0.12) a	2.6	3 (0.13)	m.
GIM	0.13	(0.05) a	0.1	3 (0.05) a	0.0	0.0/	d (†	0.13	(0.05)	a	0.09	(0.04)	q	0.16	(0.05)	ъ	0.11	(0.05) b	0.0	9 (0.04)	0
%Affiliated %ESOP %Director	-1.87 -2.66 -1.61	(0.99) b (1.81) c (1.87)																			
%Officer %Outsider	-3.74 -2.11	(2.27) b (0.47) a																			
Affiliated			-0.7	1 (0.26) a	-0.4	4 (0.27	2) C	-0.34	(0.25)	U	-0.67	(0.27)	в	-0.48	(0.29)	q	-0.63	(0.28) b	-0.6	0 (0.27)	~
ESOP			0.0	5 (0.23)	0.1	0 (0.23	ŝ	0.04	(0.25)		-0.07	(0.23)		-0.09	(0.23)		-0.14	(0.24)	0.0	7 (0.23)	
Director			0. o	3 (0.27)	F. 0 0	0.27		0.12	(0.27)		-0.01	(0.27)		0.08	(0.27)		-0.10	(0.26)	0.0	2 (0.27)	
Outsider			0.0- 0.0-	5 (0.13) a	-	0.16 0.16	0) D	-0.16	(0.17)	U	00.0-	(07.0)	ن د	2.0	(07.0)		0.40	(171) C	0.0-	(07.0) 0	
Blockholding aggregated					4	-9 C) Pi	2	-F 60	(5.73)	2	-7 01	(1 87)	a	-7 49	(1 01)	a	-5 8 1	Ч (296)	46	(20.6)	
%Medium					-2.8	5 (1.0 ⁶	a 6	-3.31	(1.88)	م د	-3.35	(0.77)	សរ	-2.14	(1.89)	5	-1.20	(1.94)	-0.8	2 (1.20)	,
%Large					-2.0	14 (0.6-	1) a	-0.97	(0.75)	U	-2.33	(0.49)	b	-0.98	(0.85)		-0.45	(0.89)	0.0	3 (1.01)	
Herfindahl index Herfindahl*Medium Uorfindahl+Lorco								-0.03	(0.15)	2				-0.14	(0.11)	U U	-0.14	(0.14) (0.00)			
Herrindani ⁻ Large								-0.14	(0.08)	٥				-0.16	(60.09)	۵	-0.19	(n.u9) b			
<i>High shareholder rights</i> (GIM ≤ 10)*Small (GIM ≤ 10)*Medium (GIM ≤ 10)*Large																	-0.29 -0.31 -0.23	(0.24) (0.21) c (0.24)			
<i>Hertindahl index</i> Hertindahl'Largest _{small} Hertindahl'Largest _{Medum} Hertindahl'Largest, _{Medum}																			-0.2 -0.2	2 (0.09) 7 (0.11) 8 (0.16)	
α _ι	6.08	(0.21) a	6.1	4 (0.19) а	6.0	9 (0.18	3) a	6.11	(0.21)	ъ	6.08	(0.18)	ъ	6.14	(0.29)	ъ	6.39	(0.20) a	6.0	3 (0.18)	
LR index (Pseudo-R ²)	0.43		0.4	e	0.4	ę		0.43			0.43			0.43			0.43		0.4	m	
Ľ	3315		331	2	331	5		3315			3315			3315			3315		331	10	
firms	826		82	6	82	9		826			826			826			826		82	6	
This table presents rando The control variables use	m effect d are: a (unbalance dummy eq	ed panel re ual to one	egressions c if the comp	f Standaro anv is a fir	d and Po	oor's cor nstitutior	porate cri	edit ratin	gs on	governa or a utilit	nce an	d contro	l variabl	les for 8	326 U.S	S. comp Market	anies over	the perio	d 1996-20	<u> </u>

ownership variables at the left hand side of the table relate to all blockholders, ownership variables at the right hand side of the table consider outside blockholders only. av denotes the estimated standard level. The Herfindahl variable measures ownership concentration, considering the (potentially) 5 largest blockholders. Herfindahl = [(%Block1) + (%Block2)]²/[(%Block1)² + (%Block2)² + ... structure, in terms of blockholding. Blockholders are shareholders owning 5 percent or more of a firm's stock. Ownership variables enter either as percentiles, always preceded by %, or as 0-1 indicators. outside blockholders (Outsider). %Small, %Medium and %Large denote the total percentage of shares held by blockholders subdivided within three brackets, with boundaries at the 5, 10 and 25 percent Margin (margin), Retained Earnings (ret), Market Beta (8), Firm Size (Size), and time dummles (ret) show). Control variables are capped at the 0.5 and 99.5 percentiles. GIM denotes the governance to one if net income before extraordinary items is negative in the previous two fiscal years (loss₍₁₋₁₎), a dummy equal to one if the firm has subordinated debt (D_{aub}), interest coverage (cov)). Operating The blockholder types considered are: Affiliated entities (Affiliated), Employee Stock Ownership Plans (ESOP), non-officer directors (Director), officers, even when they are also directors (Officer) and + (%Block5)³]. Largest_{smain}. Largest_{smain} and Largest_{arge} indicate whether the largest blockholder holds a block position in the intervals [5,10), [10,15), or 15 and beyond. Except for the type dummies, index proposed by Gompers et al. (2003), consisting of 24 provisions, which adds 1 point if a provisions compromises on shareholder rights (GIM). The remaining variables refer to a firm's ownership deviation of the random effect. McFadden's LR index (Pseudo-R³) equals 1-In(L₀), where L₀ denotes the likelihood statistic when, apart from a constant, no explanatory variables are included. Standard errors are in parentheses. a,b,c denote statistical significance at respectively the 1, 5 and 10 percent level.

Table IV: Credit Ratings - Governance Provisions and Blockholding

29